Author's response to reviews

Title: Everolimus and long acting octreotide as a volume reducing treatment of polycystic livers (ELATE): study protocol for a randomized controlled trial

Authors:

Melissa Chrispijn (M.Chrispijn@mdl.umcn.nl)
Joost PH Drenth (joostphdrenth@cs.com)

Version: 6 Date: 11 November 2011

Author's response to reviews: see over
November 11, 2011

Dear Editor-in-Chief,

Please find enclosed the revised version of our manuscript ‘Everolimus and long acting octreotide as a volume reducing treatment of polycystic livers (ELATE): study protocol for a randomized controlled trial’ (Manuscript number 2041726421576608), which we would like to contribute to Trials.

We thank the reviewer for his additional comment on the statistical section of the manuscript. We have added that we will perform additional sensitivity analysis on the missing data (p.9, l.12-13). We have marked the added part in the manuscript using a red and underlined font. Furthermore, we have changed the title into ‘Everolimus and long acting octreotide as a volume reducing treatment of polycystic livers (ELATE): study protocol for a randomized controlled trial’, following the format of Trials. We have also removed the additional files.

We thank the editor for reconsidering the manuscript and we hope we have answered to your comments sufficiently.

Joost PH Drenth, MD, Ph.D.

Professor of Molecular Gastroenterology & Hepatology
Department of Gastroenterology & Hepatology
Radboud University Nijmegen Medical Centre
P.O. Box 9101, code 455
6500HB Nijmegen
The Netherlands
Cell +31 629501892
Tel +31 24 3614760
Fax +31 24 3540103
E-mail joostphdrenth@cs.com